Author's response to reviews

Title: Using data envelopment analysis to measure the extent of technical efficiency of public health centres in Ghana

Authors:

James Akazili (akazjames@yahoo.com)
Martin Adjuik (madjuik@gmail.com)
Caroline Jehu-Appiah (carojehu@yahoo.co.uk)
Eyob Zere (zeyob@yahoo.com)

Version: 4 Date: 17 November 2008

Author's response to reviews: see over
The Editor-in-Chief  
BMC pregnancy and birth

Dear Sir/Madam,

**Resubmission of Manuscript**

I wish to re-submit my manuscript title: *Using Data Envelopment Analysis to Measure the extent of Technical Efficiency of Public Health Centres in Ghana* in your renowned journal. The issues raised in the paper are of BMC pregnancy and birth.

Revision:
1. I removed all footnotes and incorporated these in the main paper
2. I have also read through to ensure all typos are corrected
3. I have removed all the titles of the figures

I appreciate your concern in this matter.

Yours Faithfully,

James Akazili